Author's response to reviews

Title: Massive right-sided hemorrhagic pleural effusion due to pancreatitis; a case report

Authors:
Dr Mohammad Reza DR. Namazi M.D (molaa@sums.ac.ir)
Dr Ashkan DR. Mowla M.D (molaa@sums.ac.ir)

Version: 2 Date: 9 Feb 2004

Dear Editor,
I have included the reviewer's comments later and I think no further changes is needed
as all the reviewer's comment have been included.
I also have done all the formatting changes which you have requested except two:
1-both Authors are from the same institution and superscript numbers for authors are not needed.
2-as this article is case report, abstract does not need to be structured as background, methods and...
so it's own format is suitable.
all the other requested changes has been done
yours
Ashkan Mowla